Author's response to reviews

Title: Giant viable hydatid cyst of the lung: a case report

Authors:

Nagi Homesh Ghallab (nagihomesh@yahoo.com)
Ali Ali Alsabahi (nagihomesh@yahoo.com)

Version: 4 Date: 14 September 2008

Author's response to reviews:

Dear The Editor, JMCR
I included the revised manuscript with correction according to the referees comment (referee 1, because referee 2 has no comment) giving a point-by-point response to the concerns:

1- Comment to the author:
Correction of the language and grammar mistakes has been done by an English language expert.

2- Abstract:
- Line one correction (mild fever)
- Last line correction (measuring).

3- Discussion:
- On the other hand
- 95% corrected to 90%
- Please if possible keep all the figures 1, 2 and 3 (because figure 3 shows the dimensions of the cyst), but if mandatory, we can omit it.

4-Quality of written English:
- The written English quality has been corrected by expert in English language.

Thank you for your patience and I will be ready to correct any more comments from you or from the referees.

Best regards
Prof. Nagi Homesh Ghallab